Author's response to reviews

Title: Fried phenotype of frailty: cross-sectional comparison of three frailty stages on various health domains.

Authors:

Linda PM Op het Veld (linda.ophetveld@zuyd.nl)
Erik van Rossum (erik.vanrossum@zuyd.nl)
Gertrudis IJM Kempen (g.kempen@maastrichtuniversity.nl)
Henrica CW de Vet (hcw.devet@vumc.nl)
KlaasJan Hajema (KlaasJan.Hajema@ggdzl.nl)
Anna JHM Beurskens (sandra.beurskens@zuyd.nl)

Version: 4 Date: 5 June 2015

Author's response to reviews: see over
Dear editors,

Thank you for accepting our manuscript entitled “Fried phenotype of frailty: cross-sectional comparison of three frailty stages on various health domains” for publication in BMC Geriatrics.

We want to thank the editors and the reviewers for their contribution. We added the gender aspects of the analysis in the study aim at the end of the Background section. Furthermore, we included details of funding and an Abbreviations section as requested. Enclosed please find our revised manuscript.

On behalf of all authors,
Yours sincerely,
Linda Op het Veld, MSc.

Zuyd University of Applied Sciences
Faculty of Health
Centre of Research Autonomy and Participation for Persons with a Chronic Illness
Nieuw Eyckholt 300, 6419 DJ Heerlen, The Netherlands